Author's response to reviews

Title: A case of phacolytic glaucoma with anterior lens capsule disruption identified by scanning electron microscopy

Authors:

Woong-Sun Yoo (oocee@naver.com)
Byeong-Jae Kim (nadadada@hanmail.net)
In-Young Chung (in0chung@hanmail.net)
Seong-Wook Seo (stramast@naver.com)
Ji-Myong Yoo (yjm@gnu.ac.kr)
Seong-Jae Kim (oocee@hanmail.net)

Version: 5 Date: 3 November 2014

Author's response to reviews:

Dear Dr. Roberti (Referee 1)
First of all, we thank for your consideration of our paper.
We agree with your comments. We used electron microscopy not for the diagnosis of phacolytic glaucoma, but rather to investigate the mechanism underlying the condition. Phacolytic glaucoma can be diagnosed by its clinical features; therefore, electron microscopy is not needed as a diagnostic tool. However, in this report, we defined phacolytic glaucoma as cases of lens particle glaucoma as continuous spectrum of disease without differentiating between two disease entities verified with electron microscopy.
We modified the conclusion paragraph deleting SEM as diagnostic tool of phacolytic glaucoma.
This manuscript checked by native english speakers and edited as they recommended.
Thank you for your comment again.

Dear Dr. Riva (Referee 2)
We really thank for your comments.
First, we totally agree with your comments. We initially tried to perform extracapsular cataract extraction (ECCE) rather than intracapsular cataract extraction (ICCE). However, during surgery, we found zonulysis in nearly two thirds of the lens. Because of this, ICCE was needed. We added in the manuscript. And we also agree that ICCE is a traumatic procedure and can cause trauma to the lens capsule. We have added this in the manuscript as a limitation of this report. ICCE was performed with lens capsule forceps and spoon which is also added in manuscript. Finally, we deleting and editing lines form 109 to 119 which repeating in result section.
This manuscript checked by native English speakers and edited as they recommended.
Thank you for your consideration.